Author’s response to reviews

Title: Serum p53 antibody detection in patients with impaired lung function

Authors:

Manlio Mattioni (mattioni@ifo.it)
Patrizia Chinzari (pattykick1ym5tc@yahoo.it)
Silvia Soddu (soddu@ifo.it)
Lidia Strigari (strigari@ifo.it)
Vincenzo Cilenti (cilenti@ifo.it)
Eliuccia Mastropasqua (mastropasqua@ifo.it)

Version: 2 Date: 21 June 2012

Author’s response to reviews: see over
Dear Miss Marinette Lacson
Journal Editorial Office
BioMed Central

Dear Miss Marinette Lacson

Please find enclosed the revised version of the manuscript “Serum p53 antibody detection in patients with impaired lung function” for consideration for publication to *BioMed Central Cancer*, in the category of research articles.

As you requested, I stated “All participants in the study gave written informed consent” at the end of Patients subsection in the Methods section.

Thank you for your suggestion.

Best regards,

*Manlio Mattioni*

Regina Elena National Cancer Institute
Experimental Research Centre
via delle Messi d’ Oro 156, 00158 Rome, Italy
telephone number +390652662528, fax number +390652662505
e-mail address mattioni@ifo.it